Author's response to reviews

Title: Cell fusion between gastric epithelial cells and mesenchymal stem cells induces epithelial-to-mesenchymal transition and malignant transformation

Authors:

Xianghui He (humphreyhe@163.com)
Baosong Li (957963980@qq.com)
Yang Shao (sdsy2004@163.com)
Na Zhao (nasy3367@126.com)
Yiling Hsu (evanhsu@hotmail.com)
Zhixiang Zhang (zhixiang63@sina.com)
Liwei Zhu (zhuliwei0804@126.com)

Version: 2 Date: 14 October 2013

Author's response to reviews:

Dear Editor,

Thank you for holding our manuscript entitled (MS: 3227807010995263) for revising.

We revised the manuscript according the editorial request:

1) The language has been corrected by native-English speaker from Edanz.
2) The abstract section has been formatted according the journal guidelines.
3) All author’s emails were added in the title page
4) The funding statement was moved to Acknowledgements section
5) Abbreviation in the title were changed, and all abbreviations and acrnyms were defined at first uasage.

Now we submit the new version for your consideration and look forward to hearing from you.

Regards,

Xianghui He, MD, PhD

Professor of General Surgery
Department of General Surgery
Tianjin Medical University General Hospital